Author's response to reviews

Title: Frequency of alcohol use and obesity in community medicine patients

Authors:

James E Rohrer PhD (james.rohrer@ttuhsc.edu)
Barbara M Rohland MD (barbara.rohland@ttuhsc.edu)
Anne Denison RN (anne.denison@ttuhsc.edu)
Anthony Way MD (anthony.way@ttuhsc.edu)

Version: 3 Date: 14 April 2005

Author's response to reviews:

Dear Editors:

We have made the following changes to the manuscript.

Author list - The list of authors in the manuscript is exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system.

Affiliations - Each affiliation is written in full and provided in the following format: department, institute, city, country.

Each author is linked to his or her corresponding affiliation with the use of superscript numbers.

The competing interests statement has been correct.